Author's response to reviews

Title: Asthma and COPD in primary health care, quality according to national guidelines: a cross-sectional and a retrospective study.

Authors:
   Siw Carlfjord (siw.carlfjord@ihs.liu.se)
   Malou Lindberg (malou.lindberg@lio.se)

Version: 5 Date: 12 June 2008

Author's response to reviews: see over
To the Editor, BMC Family Practice,

We hereby submit our revised manuscript “Asthma and COPD in primary health care, quality according to national guidelines: a cross-sectional and a retrospective study”.

We are pleased that the manuscript has been accepted for publication in BMC Family Practice.

All formatting changes requested are performed. We also made a clarification in the manuscript in the paragraph “Correlations between structure and process” line 5, adding the text “in the asthma group”.

Yours sincerely

Siw Carlfjord
R.P.T., M.P.H.